

# understanding micro drivers of the aggregate effect Natália Barbosa | Ana Paula Faria

Gabinete de Estratégia e Estudos da Economia e do Mar Office for Strategy and Studies of Economy and Maritime Affairs Avenida da República, n.º 79 - 1069-218 Lisboa, Portugal <u>www.gee.gov.pt</u> ISSN (online): 1647-6212









## Digital adoption and productivity: understanding micro drivers of the aggregate effect <sup>1</sup>

Natália Barbosa<sup>2</sup>, Ana Paula Faria<sup>3</sup>

#### Abstract

Digital technologies have the scope to engender positive effects on productivity at firm and aggregate level. However, empirical evidence and theoretical contributions are ambiguous as mixed findings and diverse explanations have been put forward. We use a rich and representative sample of Portuguese firms over the period 2014-2019 to empirically assess the relationship between digital technologies adoption and productivity. Based on estimations over the entire distribution of firm's productivity, we find that heterogeneous digital technologies affect differently the dynamics of productivity and the convergence to the frontier. This leads to mixed findings with scope to diverse impact in the aggregate productivity. Moreover, positive and significant effects on productivity require an upgrading on the degree of sophistication and complementarity among digital technologies and benefit from the ability of firms to interact and learn with digitalised peers in the same industry.

**Keywords:** Digital technologies, Productivity, Spillover effects **JEL Classification:** L20, H81, L25

# Note: This article is sole responsibility of the authors and do not necessarily reflect the positions of GEE or the Portuguese Ministry of Economy and Maritime Affairs.

<sup>&</sup>lt;sup>1</sup> Award-winning work by the Call for Papers on The Economic Impact of Digitalisation in Portugal, a partnership between the Office for Strategy and Studies of the Ministry of Economy and Maritime Affairs (GEE), Google and the Association for the Information Society Development (APDSI).

<sup>&</sup>lt;sup>2</sup> Department of Economics and NIPE, University of Minho, Campus de Gualtar, 4710-057 Braga, Portugal. E-mail: natbar@eeg.uminho.pt

<sup>&</sup>lt;sup>3</sup> Department of Economics and NIPE, University of Minho, Campus de Gualtar, 4710-057 Braga, Portugal. E-mail: apfaria@eeg.uminho.pt



### 1. Introduction

Significant scholarly attention has recently been devoted to understanding the evolution of aggregate productivity, particularly in developed countries, as it has been showing a declining trend, despite significant technological changes, capital investment and various public programs supporting firms. In a context of slow global productivity growth, several studies have led to renewed discussions about Robert Solow's 1987 productivity paradox. These studies suggest that the effects of recent technological changes, and similarly of various programs to support firms, tend not to be visible in terms of aggregate productivity. Some contributions, for example, Pinheiro Alves (2017) for the case of the Portuguese economy, were carried out to document and explain the phenomenon. Additionally, since the beginning of the 2000s, there has been a decrease in business dynamics in several countries (Criscuolo et al., 2014), which indicates constraints in terms of resources reallocation and, consequently, in obtaining productivity gains.

Although the innovation effort by the most innovative firms did not register significant decreases, Andrews et al. (2015) suggest that a possible explanation for the absence of visible effects in terms of aggregate productivity could be a decrease in the rate at which innovations spread and replicate across the economy. Andrews et al. (2015) named it as a break in the technological diffusion machine. The ability of other firms to learn from more innovative frontier firms may have decreased, which is consistent with previous evidence in terms of penetration rate of new technologies (e.g., Comin and Mestieri, 2013) or even the possibility of innovators to appropriate all the dynamics and benefits arising from innovation (Gabaix and Landier, 2008). However, other authors argue that the growing digitization of economic activities may reactivate the technological diffusion machine, as it makes business trialling faster and cheaper and the possibility of replicating innovations faster and more accurate (Brynjolfsson and McAfee, 2012). If so, positive effects in terms of firm level and aggregate productivity can be expected as digital technologies adoption spreads out across firms and industries.

In fact, some authors argue that there are good reasons to believe that digital technologies should have the power to engender strong positive effects on productivity (Syverson, 2011; Brynjolfsson and McAfee, 2014). Innovation is how the productivity growth happens and many innovations emerge from recombining existing solutions and elements in order to make them more valuable. Digital technologies assist this kind of innovation by making it easier to combine ideas. When firms began to make widespread efforts to integrate digital technologies into their business, for example by improving business processes, automating certain routine tasks, and reducing costs of interacting with suppliers and customers, productivity gains are likely to emerge (Bartel et al., 2007; Akerman et al., 2015).

Yet, the empirical evidence at the industry and firm levels has yield mixed findings with scope to diverse explanations and many directions of further research (see, for instance, Acemoglu et al., 2014; Bartelsman et al., 2017; DeStefano et al., 2018). Andrews et al. (2016) and Berlingieri et al. (2020) have shown that aggregate patterns mask a widening productivity





gap between a handful of frontier firms and a mass of laggard firms (firms that operate at the bottom of productivity distribution), especially in highly digitalized industries. Moreover, digital technologies tend to engender productivity gains only in combination with other factors such as organisational and management skills (e.g., Bloom et al., 2012), R&D and intangible investments (e.g., Corrado et al., 2017) and regulatory environment that enables the efficient resources reallocation (e.g., Bartelsman, 2013). In addition, digital technologies are a heterogeneous set of technologies that may affect firms' productivity in different ways. Therefore, previous evidence points out that the links between digital adoption and productivity are complex (for a survey, see, e.g., Gal et al., 2019) and deserve further empirical research.

The objective of this paper is three-fold. Firstly, we take a broader perspective and assess how the adoption of a range of digital technologies could affect productivity at firm level for a representative sample of Portuguese firms over the period 2014-2019. The digital technologies under scrutiny are a set o heterogenous technologies and have been selected for their potential to improve firm productivity and for being in different maturity stages. Second, we explore some of the reasons why the benefits at firm level could be disappointing at the aggregate level. In particular, we explore whether there are differences in the productivity benefits from digital technologies adoption among firms belonging to different industries and with different levels of productivity and whether digital adoption is able to foster productivity convergence between frontier and laggard firms. Third, another important issue in understanding the link between digital adoption and productivity is related to benefits that could be derived from interactions with digitalised peers in the same industry. This implies to rely into detailed firm level and to define an empirical framework that helps in disentangling whether productivity benefits are driven by within-firm adoption or from spillovers due to interactions with other digitalised firms in the same industry.

As far as we are aware, there is no similar study applied to Portuguese firms. Assessing the impact of digitalisation on firm-level productivity and the way it translates into productivity growth at aggregate level is crucial and could provide important implications for public policy related to the support of digital adoption and the creation of conditions that enable a catch-up process of laggard firms. Moreover, to date, there is only some initial evidence that the latest generation of new technologies, namely big data and highly automated production modes, have positive effects on productivity. Furthermore, by analysing technologies in different maturity stages, we provide evidence to the extent productivity is following a J-curve shape (Brynjolfsson et al., 2021). Therefore, our study provides a contribution to this new strand of literature, analysing those effects associated with the adoption of such transformative technologies in a developed country but, in the context of the European Union (EU), a small, open, and peripheral economy.

The paper is organised as follows. Section 2 presents an overview on the links between digital technologies and productivity. Section 3 describes the data, empirical variables, and provides a description of digital technologies adoption by Portuguese firms. Section 4 describes





the empirical strategy. Subsequently, econometric results are presented and discussed in Section 5. Section 6 concludes and discusses policy implications.

#### 2. Digital technologies and firm's productivity

It has been widely recognised by scholars that innovation is a crucial driver of productivity increase. At the firm-level, innovation may lead to productivity increases through exploitation of scale economies, efficiency change, technological change and also through the use of unused capacity utilisation. The vast extant literature on the topic has shown that innovation may affect productivity in different ways depending on its type – product, process or organizsational, and novelty – incremental and/or radical (Mohnen and Hall, 2013).

Process innovations in the form of new technologies – such as digital technologies, are seen as powerful productivity drivers. Digital technologies, also called intelligent or smart technologies (Brynjolfsson and McAfee, 2012), are seen as envisaging the Fourth Industrial Revolution. They are particularly important because they are an example of general purpose technologies (GPTs), that is, technologies that are introduced very infrequently but they can have a significant impact on firms and on the productivity of an economy (Bresnahan and Trajtenberg, 1995; Jovanovic and Rousseau, 2005). This is because they have both the capacity to change the ways in which firms conduct business and a pervasive nature, i.e., they can be applied in many ways and areas with far-reaching economic and social consequences (Bresnahan and Trajtenberg, 1995).

Despite the potential positive impact of digital technologies on firms' productivity, their adoption is not instantaneous. It is an empirical regularity that the diffusion of new technologies is a process that takes time, typically described by a sigmoid curve (Stoneman, 2001). The technology diffusion literature shows that a firm decision to adopt process technologies is determined by a set of factors relating the technology (e.g. novelty, complexity) and firm-specific characteristics (e.g. resources and knowledge endowments), as well as the external context in which the firm operates (e.g. competition, number of current adopters, different demand conditions, level and cumulativeness of knowledge (Malerba and Orsenigo, 1996). Thereby, it is an empirical regularity that as technology diffusion takes place one observes differences in the adoption rates across firms, which in turn leads to firm productivity heterogeneity (Hall et al., 2009).

In the case of digital technologies adoption is expected to take time due a number of aspects. The radicalness of technologies usually implies a destruction of routines as the new technology may not be compatible with the firms existing production lines or with old technologies embedded in old capital (Hobijn and Jovanovic, 2001). Also, the adoption of these technologies requires complementary assets such as management practices, skill-specific labour or the internal organization adjustments necessary to deploy digital technologies successfully (Bresnahan et al., 2002; Bloom et al., 2012). So, the general view is that digital technologies require an initial learning period, during which the firm becomes





more productive through the efficient use of these technologies. This implies that productivity growth tends to decline in the early stages of the diffusion of the new technology, which has been described as a productivity J-curve (Brynjolfsson et al., 2021).

Recent studies on digital technologies adoption and their impact on firm productivity have been recognising that digital technologies are not homogenous and should be treated differently (Syverson, 2017; DeStefano et al., 2018; Ballestar et al., 2020). Digital technologies relate to various technologies where the most preeminent are enterprise resource planning (ERP), customer relationship management (CRM), and enterprise content management (ECM), cloud computing, big data and automation, each with different capabilities (Frank et al., 2019). ERP and CRM technologies are mature software systems and data technologies, which lack more sophisticated features encompassed by big data analytics and automation. Likewise, recent developments in industrial robotics allowing its connection between artificial intelligence and the second-digital-wave technologies (i.e., machinelearning, internet of things, cloud computing, big data, or 3D printing) have brought robotics into a new level of technological sophistication, generating renewed academic interest in their effects on productivity and employment.

Empirical evidence on digital technologies adoption and impact on productivity has almost exclusively focused on their first wave (i.e., ICT relating to internet use and ERP and CRM technologies), but do not discriminate among different technologies (e.g., Corrado et al., 2017; Chun et al., 2015; Pieri et al., 2018; Stiroh, 2002). Overall, this research shows a positive impact of ICT on firm productivity through improvements in efficiency at the firm level. These contributions also find that the impact of ICT investments on productivity growth is higher in sectors that produce or intensively use ICT capital goods and when complemented with other internal capabilities.

More recent studies have investigated the impact of set of latest generation of new technologies on productivity, namely robots (Ballestar et al., 2020; Jungmittag, 2021) and artificial intelligence, big data, flexible automation (Venturini, 2021). They essentially corroborate the evidence found in previous generations of technologies, in that, they find a positive link between these technologies and productivity. In addition, their evidence corroborates the hypothesis of the productivity J – curve, as initially these technologies increase productivity dispersion and only after the initial phase some convergence is observed.

In addition to the within-firm effect of digital technologies-driven productivity dispersion among firms, some contributions (Chun et al., 2015; Corrado et al., 2017; Pieri et al., 2018) have found that between-firms effects are equally important to explain aggregate productivity. Specifically, Chun et al. (2015) show there are significant cross-industry variations in firmlevel resource reallocation from less productive firms to more productive ones. Thereby, industries in which the reallocation mechanism is more efficient will be able to attain higher productivity levels. More recently, Corrado et al. (2017) and Pieri et al. (2018) add to these findings and find that digital technologies alongside R&D have been particularly effective in





generating inter-industry spillovers that contribute to productivity growth. The spillovers effect of digital technologies appears to be stronger in high-tech sectors. Audretsch and Belitski (2020) have found corroborating evidence of the importance of knowledge spillovers associated with new technologies to explain firm productivity, highlighting the importance of the phenomenon.

#### 3. Data and empirical variables

In this paper we rely on two main panel datasets provided by the Portuguese National Institute of Statistic (INE): *Inquérito à utilização de tecnologias de informação e comunicação nas empresas* (IUTIC) for digital adoption and *Sistema de Contas Integradas das Empresas* (SCIE for firm-level productivity and other firm-specific characteristics. IUTIC is part of the Community Survey on ICT Usage and E-Commerce and is an annually survey conducted since 2004. In Portugal, this survey is a census for large firms (with more than 250 employees or turnover larger than 25 million euros) and a stratified random sample based on size and industry affiliation for other firms. The survey is compulsory by law for selected firms, which makes it a reliable, rich and valuable dataset. The survey encompasses several questions related to e-commerce and developments in digital technologies' usage in enterprises that allow us to assess the adoption of digital technologies in transforming the business world.

Looking at digital technologies, the survey asks several questions that goes from a simple internet usage to a more sophisticated and intensive usage of digital technologies, such as cloud computing, robots, big data collected from digital social media, sensors or other digital devices, or digital technologies to share information among firm's departments and customers. We interpret the answer to these questions as indicating that a firm adopts digital technology that could affect productivity. In particular, to our purpose, we focus on three digital technologies: ERP (Enterprise Resource Planning) and CRM (Customer Relationship Management) technologies - which are firm-wide and inter-firm system applications-, Big Data, and use of robots. These questions are intermittently available since 2014. Therefore, the time span of our main analysis is the period 2014-2019, even though with the gaps for different variables. Table 1 describe the intensity of usage of the analysed digital technologies, along with the diferences among specific types of firms. We consider firm's size, technological and digital intensity at industry level as the criteria to identify types of firms. The criterion to identify firms belonging to high-tech industry was based on (Galindo-Rueda and Verger, 2016). In turn, we consider industries as digital-intensive if they were classified by Calvino et al. (2018) as medium-high or high digital intensive for the period 2013-2015.





	(1)	(2)	(3)	(4)	(5)
	Ν	mean	sd	min	max
PANEL A: All firms					
Adoption of ERP and CRM technologies	24,417	0.513	0.500	0	1
Adoption of Big data analysis and usage	18,134	0.141	0.348	0	1
Adoption of robotic technologies	12,519	0.092	0.289	0	1
PANEL B: Large firms					
Adoption of ERP and CRM technologies	866	0.932	0.252	0	1
Adoption of Big data analysis and usage	772	0.323	0.468	0	1
Adoption of robotic technologies	759	0.298	0.458	0	1
PANEL C: SME firms					
Adoption of ERP and CRM technologies	11,966	0.705	0.456	0	1
Adoption of Big data analysis and usage	9,064	0.186	0.389	0	1
Adoption of robotic technologies	5,725	0.139	0.346	0	1
PANEL D: Firms in high-tech industries					
Adoption of ERP and CRM technologies	1,870	0.620	0.485	0	1
Adoption of Big data analysis and usage	954	0.237	0.425	0	1
Adoption of robotic technologies	502	0.062	0.241	0	1
PANEL E: Firms in high-digital inte	nsity				
Adoption of ERP and CRM technologies	4,846	0.577	0.494	0	1
Adoption of Big data analysis and usage	2,495	0.155	0.362	0	1
Adoption of robotic technologies	1,241	0.062	0.241	0	1

Table 1: Distribution of digital technologies across firms

All firms report the adoption of cloud computing, which suggest that they are aware of the need and the benefits extracted from digital technologies. Therefore, cloud computing could not be seen as having predictive power in explaining differentials in firms' productivity gains or performance and, hence it is not included in this study. More interestingly, Portuguese firms seem to follow a heterogenous behaviour in terms of adoption of digital technologies. ERP and CRM technologies are adopted for more than 50% of firms, regardless of the type of firms one could consider. It is a digital technology largely spread across firms, which also indicate some maturity in its usage.

Conversely, the adoption of robotic technologies is a more recente phenomena, with a small percentage of adopters. Less than 10% of firms report having adopted such digital technology. Comparing with other European countries the difference is even more striking. In 2015, the adoption rate of robots was 22% in Spain (Ballestar et al., 2020) and 64% in Germany (Jungmittag, 2021). And, interestingly, firms belonging to hightech or high-digital intensity industries seem not be those more prone to adopt robotic technologies, suggesting that the drivers for adoption of these technologies is somewhat unrelated to digital and technological intensity at industry level. To some extent, this low diffusion of robots in Portuguese firms reflects the countrys industry specialization in sectors that are less prone to adopt robots, like wood and paper, textiles or even chemicals. Nonetheless, another possible reason is because the investment required to robotic technologies is quite capital demanding that only some firms are able to do so. In particular, large firms, irrespectively the industry where they belong, are those more prone to adopt robotic technologies. Moreover, large firms appear to be the big adopters of digital technologies, suggesting that firm's size could be an important driver of digital technology adoption.





Looking at Big Data analysis and usage, apart from firm's size, technological intensity at industry-level appears to foster the adoption of this digital technology. Firms belonging to high-tech industries seem to attach more value to the digital transformation associated to the analysis and use of big data, as a way to empower them with differentiable capabilities to reach markets, adjust products and process to customers needs, and to obtain performance gains. Comparatively, the analysis and use of big data seem to be the distinctive feature of high-tech firms with respect to digital technological adoption and transformation.

We then match IUTIC firm-level data with data from SCIE, which is an annual census for Portuguese firms, encompassing a rich set of firm-level information such as gross output, added value, capital stock, employment and workforce characteristics, export, and the industry in which firms operate according to the NACE classification (Classification of Economic Activities in the European Union). The SCIE database also provide information on the number of employees developing research activities and firms' sales in domestic and external markets, allowing us to compute firm's export intensity and R&D intensity. This set of information is the basis for computing variables affecting productivity performance such as firm size, technological opportunities, and export orientation (Bartelsman and Doms, 2000), apart from our variables of interest related to digital technologies adoption.

As both datasets include the same unique firm identifiers, we are able to trace firms over time and conduct the empirical assessment. We match 26,018 unique firms for which we have intermittently information on digital technologies adoption. This intermittency is due to variations on the IUTIC survey design and sampling scheme. Therefore, our final sample consists of 39,411 firm-year observations, where 5,909 firms are observed in at least two years. Table A.1 in Appendix A describe the measurement and source of the empirical variables used in the analysis, whereas Table 2 reports some descriptive statistics for dependent and control variables, averaged over all firm-year observations and over several different partitions of the data. For each partition, the last column shows the statistical difference (given by a t-test) of the means of these variables for the types of firms identified.





	All firms		S	SMEs		High-Tech		-Digital
	Mean	Sd.	Mean	Diff.	Mean	Diff.	Mean	Diff.
Dependent variables								
Labour Productivity	10.14	1.058	10.35	0.409***	10.40	0.278***	10.14	0.004
Turnover	14.15	2.502	15.87	3.254***	13.91	-0.264***	13.934	-0.266***
Distance to productivity frontier	0.921	0.079	0.935	0.029***	0.915	-0.006***	0.914	-0.008***
Control variables								
Intensity of ICT	0.025	0.108	0.010	0.030***	0.095	0.076***	0.062	0.047***
Firm's size	14.09	2.534	15.688	3.042***	14.00	-0.105**	13.82	-0.342***
Exporting intensity	0.090	0.232	0.148	0.112***	0.065	-0.026***	0.056	-0.041***
Financial constraints	1.073	13.47	0.676	-0.760***	1.165	0.100	0.925	-0.184
Intensity of R&D	0.006	0.050	0.008	0.005***	0.024	0.019***	0.008	0.003***

#### Table 2: Descriptive statistics of dependent and control variables

Notes: \*\*\*. \*\*, \* indicate that the difference in means is statistically significant at the 1%, 5%, 10%, respectively.

Overall, Portuguese firms report a very low intensity on R&D and a low share of workers involved only in ICT activities, suggesting that their performance and efficiency could be strongly compromised. In particular, on average, the low share of workers involved only in ICT activities could be interpreted as indicating that the process of digital technologies adoption deepness is still at the beginning, with, potentially, a low degree of sophistication and combinations of use. Portuguese firms seemingly are more prone to adopt digital technologies that could be supported by external consulting or assistance, rather than more sophisticated digital technologies that require a permanent assistance and in-house improvements in their use. Nonetheless, firms belonging to hightech or digital intensive industries have, on average, higher shares of workers involved only in ICT activities, suggesting that a deeper and longer involvement on digital technologies adoption and greater ICT-based capabilities. Instead, SMEs appear to be much less intensive in ICT workers, which could be interpreted as being at the initial phase of digital technologies adoption process.

On the other hand, Portuguese firms reveal strong financial constraints as the reported liabilities are, on average, greater than total assets. This could be a relevant constraint for the adoption process and the way firms are able to extract performance gains from digital technologies. Interestingly, the nature of industries - high-tech or digital intensive - seems not impose, comparatively, additional financial constraints. The little involvement in international markets, and hence their reduced exposure to more competitive markets, could be seen as an indictor of a fragility to improve performance. Only 30.4% of the firms are exporters and, for them, the average of exporting intensity does not exceed 30%, indicating that there is a large scope to expand activity in international markets and, potentially, performance. More importantly, firms belonging to high-tech or digital intensive industries have a smaller exposure to international - more competitive - markets, suggesting some difficulty to overcome barrier in operating in these markets with statistically significant and negative impact on turnover and on the ability to shorten the gap to frontier firms.



These heterogeneity in terms of firm's resources and competence could be seen as having predictive power for the heterogenous performance, measured by labour productivity and turnover, and as moderate factors for the relationship between performance and digital technologies adoption.

### 4. Empirical strategy

To examine the relationship between firms' productivity and digital technologies adoption across the whole conditional distribution and to allow for the possibility of the existence of asymmetric dynamics, we apply quantile regression models as they are able to describe that relationship at different points in the conditional distribution of firm's productivity. It is a powerful tool for comparing, more thoroughly than the mean, various relevant points of a distribution across different patterns of variables of interest and other covariates. And, hence, it may show whether the variables of interest related to digital technologies exerts a significant influence on one tail of the distribution but not on the other. In particular, it allows us to asses whether a productivity gap between laggard and frontier firms is being widened or shortened.

By modelling the conditional quantiles,  $q \in (0,1)$  of the form  $Q_q(Y|x,z) = x'_q \alpha_q + z'_q \beta_q$  where Y is the alternative measures of firm's productivity, x is a vetor of variables related to digital technologies adoption and z is a vector of control variables that account for differences in several observable attributes of the firm. Although its computation requires linear programming methods, the quantile regression estimator is asymptotically normally distributed, more robust to outliers than the OLS regression, and its semi-parametric nature avoids assumptions about the parametric distribution of the error process (Koenker, 2005).

Estimation was performed using the SQREG procedure in STATA and a bootstrap resampling approach, based on 100 replications, were performed to estimate the entire variance-covariance matrix of the estimators. The relevance of the bootstrapping procedure hinges on its robustness property; in particular when the errors from the quantile equation are not homogeneously distributed. Another important point is that the full sample is used for every quantile regression and not only the observations belonging to marginal quantiles (i.e. the raw quantiles of the *Y* function). This implies that all covariates are valid predictors in all quantiles.

Other important feature of our empirical strategy concerns the time-lagged variables in the model specification. Here, the short length of our unbalanced panel data, due to the frequent changes on the IUTIC survey questions related to digital technologies adoption over the years, prevent us to directly specify a time-lag for the covariates. This would imply a drastic loss of sampled firms and observations. Nonetheless, the IUTIC survey variables reflect firm's status at the beginning of each year, whereas the productivity and control variables were measured at the end of each year. Thus, in practice, our data already consider a certain time lag between our main independent variables and the alternative measures of productivity. Therefore, our





specification, in certain sense, considers that the effect of digital technologies adoption could take time to affect firm's productivity.

On the other hand, to control to potential spurious correlations between productivity and adoption of digital technologies due to, for instance, demand shocks that affect both the choice of adoption and firm productivity or technological progress that leads firms to invest in digital technologies and, simultaneously, in other productivity-enhancing activities, our specification allow for industry-specific effects in order to control for these potentially confounding trends at industry-level.

#### 5. Impact of digital technologies adoption and productivity

Here, the empirical results related to the relationship between digital technologies adoption are presented. In the first stage, empirical results based on all firms in our sample are presented and discussed. Our focus is on three digital technologies - ERP and CRM, Big Data, and Robots - and firm's labour productivity. Moreover, each empirical model considers a direct impact of a single adoption of a digital technology, an effect related to the degree of sophistication and combinations of use of different digital technologies - proxied by the share of workers involved only in ICT activities ar firm level -, and an interaction term of both as a way to measuring the impact of the complementarity of digital strategic resources.

In the second stage, the robustness of the results is addressed and discussed. For that, the alternative models are re-estimated using several different partitions of the data, which allows us to uncover potentially heterogeneous relationships between digital technologies adoption and productivity and, hence, to understanding the aggregate effect. These partitions are based on firm's size, high-tech industries based on Galindo-Rueda and Verger (2016) and digital-intensive industries based on Calvino et al. (2018).

#### 5.1 All firms

Although firms may adopt some digital technologies for purposes that do not target productivity growth directly (e.g. investments that aim simply to substitute an existing technology by a new up-to-date solution without engendering noticeable changes on products and production), they do expect qualitative impacts, such as better sharing and use of inhouse information that would generate performance improvements indirectly. In this sense, any investment in digital technologies ultimately is expected to engender performance gains. Table 3 shows the estimated impact of different digital technologies over the distribution of labour productivity at firm-level.





#### Table 3: Impact of digital technologies on labour productivity

			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	0.009	0.007	-0.037***	-0.093***	-0.183***
	(0.023)	(0.012)	(0.012)	(0.016)	(0.023)
Intensity of ICT	0.230*	0.428***	0.558***	0.750***	0.573***
	(0.130)	(0.107)	(0.080)	(0.103)	(0.063)
Intensity of ICT x Adoption of ERP and CRM technologies	0.410***	0.319**	0.556***	0.972***	1.746***
	(0.150)	(0.139)	(0.131)	(0.212)	(0.376)
Control variables					
Firm's size	0.149***	0.168***	0.205***	0.267***	0.344***
	(0.005)	(0.003)	(0.003)	(0.004)	(0.005)
Exporting intensity	-0.043	-0.054**	-0.079***	-0.156***	-0.201***
	(0.031)	(0.022)	(0.021)	(0.031)	(0.052)
Financial constraints	0.330***	-0.158***	-0.066***	-0.017**	0.003
	(0.030)	(0.021)	(0.011)	(0.007)	(0.008)
Intensity of R&D	0.357	0.432**	0.527***	0.240**	0.027
	(0.236)	(0.218)	(0.086)	(0.117)	(0.159)
Manufacturing firms	0.114***	-0.046***	-0.176***	-0.318***	-0.484***
5	(0.019)	(0.011)	(0.010)	(0.015)	(0.020)
Wholesale and retail trade firms	0.144***	0.054***	0.005	0.008	-0.149***
	(0.020)	(0.013)	(0.015)	(0.022)	(0.023)
Constant	7.394***	7.456***	7.315***	6.903***	6.427***
	(0.075)	(0.044)	(0.038)	(0.054)	(0.063)
Pseudo R2	0.15	0.18	0.20	0.23	0.27
N	23,070	23,070	23,070	23,070	23,070
Panel B: Big Data Technologies		/			
Adoption of Big data analysis and usage	0.019	-0.007	-0.047***	-0.119***	-0.192***
, , ,	(0.030)	(0.019)	(0.017)	(0.024)	(0.035)
Intensity of ICT	0.291*	0.506***	0.795***	0.873***	0.911***
	(0.169)	(0.113)	(0.089)	(0.064)	(0.156)
Intensity of ICT x Adoption of Big Data	-0.021	0.121	0.064	1.568	3.405***
	(0.728)	(0.251)	(0.306)	(1.167)	(0.633)
Pseudo R2	0.15	0.18	0.21	0.23	0.27
N	11,842	11,842	11,842	11,842	11,842
Panel C: Robotic Technologies		,		/= · _	
Adoption of robotic technologies	0.073	0.016	-0.055*	-0.209***	-0.338***
	(0.046)	(0.033)	(0.032)	(0.048)	(0.056)
Intensity of ICT	0.390	0.549***	0.784***	0.876***	1.232***
	(0.266)	(0.172)	(0.078)	(0.141)	(0.433)
Intensity of ICT x Adoption of robots	0.222	0.640	0.829	0.971	1.623
	(1.461)	(1.101)	(0.958)	(1.215)	(1.602)
Pseudo R2	0.14	0.17	0.21	0.23	0.26

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*,

1%. All models included the control variables showed on Panel A.

Overall, productivity improvements seem to depend critically on how the digital technology adopted is combined with in-house ICT capabilities, disclosing the role of the degree of sophistication and complementarities among digital technologies and other resources in extracting productivity gains. This result is a well-documented feature in the literature (Brynjolfsson et al., 2021; Chun et al., 2015; DeStefano et al., 2018; Pieri et al., 2018). Regardless of the firm position at the distribution of labour productivity, an increase in ICT





capabilities - proxied by the share of employees involved only in ICT activities - and an upgrading in ICT use sophistication (Miyazaki et al., 2012), would yield productivity gains. This positive and statistical effect seems to increase with firm's productivity level, being frontier firms the bigger extractors of productivity benefits. Moreover, the adoption of digital technologies without in-house ICT capabilities, which can be interpreted as an initial (or, takeoff) phase of digital technology adoption, seems to engender a negative impact on productivity for firms at the higher quantiles of the productivity distribution (i.e, greater performers and frontier firms). This would imply that initial investments on digital technologies, not complemented with ICT capabilities, could require an learning and adapting process, changes on management practices and decision-processes that appear to impact negatively on production efficiency. Ignoring these complementarities may introduce bias into the estimated effect of digital technologies on productivity. This finding is in line with the productivity J-curve argument developed by Brynjolfsson et al. (2021), in which the need for intangible investments, such as investments in creating new business processes or training workers, in early stages of new general purpose technologies adoption - as it is the case of digital technologies - would cause an initial productivity slowdown.

Further, looking at specific digital technologies, enterprise resource planning (ERP) software and customer relationship management (CRM) technologies, aimed to simplify and centralize business processes and to gain better insights into their customers to make more informed decisions, complemented with in-house ICT capabilities, seems to engender productivity gains for all firms across the productivity distribution. Nonetheless, frontier firms are those that are able to extract greater benefits from that digital technology, suggesting that digitalisation based on ERP and CRM technologies are more likely to exacerbate existing divides, rather than closing productivity gaps. Moreover, these productivity gaps seem to be further exacerbated by the better-informed decision-making processes based on novel data practices related to big data technology. In fact, only frontier firms exploring big data are able to extract productivity gains, indicating that the benefits of big data are still more scare and potentially require the previous deployment of other productivity-enhancing capabilities. Nonetheless, and similarly to Niebel et al. (2019), our findings provide evidence on the economic value of big data. But not all firms that adopt such digital technology exhibit productivity gains. The complementarity of other resources and capabilities could be an explanation for the estimated differential impact.

A more challenging finding is that related to robotic technologies adoption. The empirical results suggest that Portuguese firms investing in robotic technologies are still not able to extract productivity gains, even when combined with different degrees of ICT intensity and sophistication. Nonetheless, the positive, although statistically insignificant, estimates for adopters with in-house ICT capabilities can be interpreted as an indication that there may be scope to extract productivity gains from robotic technologies. These results also suggest that a digitalisation strategy based on robotic technologies may require a longer time span to





engender productivity gains. These digital technologies tend to be longer duration investments that could prolong the effects of the J-curve.

On the side of other variables affecting productivity gains, some interesting findings emerge. Apart from the stylised facts that firm's size and R&D have predictive power in explaining productivity differentials, financial constraints emerge as a important factor in shaping productivity gains. In particular, only frontier firms seem to escape to this tie, as productivity gains of them are not affected by financial constraints. To other firms, greater financial constraints seem to lead to an increasing productivity gap. Moreover, looking at laggard firms, those operating into the manufacturing or wholesale and retail trade sectors seems to perform better than other firms, suggesting that they are more likely to climb the productivity distribution.

Another important perspective is to assess whether digital technologies have an important role in closing the productivity gap to frontier firms. Table 4 shows the estimated impacts of different digital technologies over the distribution of the distance to frontier firms, i.e. the greatest performers at industry-year level based on labour productivity.

			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	0.006**	-0.001	-0.003***	-0.007***	-0.009***
	(0.002)	(0.002)	(0.001)	(0.001)	(0.001)
Intensity of ICT	0.006	0.003	0.002	0.004	0.015**
	(0.008)	(0.009)	(0.005)	(0.010)	(0.007)
Intensity of ICT x Adoption of ERP and CRM tecnologies	0.012	0.011	0.015	0.037***	0.062***
	(0.014)	(0.014)	(0.012)	(0.013)	(0.015)
Control variables					
Firm's size	0.006***	0.008***	0.009***	0.011***	0.013***
	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)
Exporting intensity	0.007	0.003	0.005***	0.005**	0.012***
	(0.004)	(0.002)	(0.002)	(0.002)	(0.004)
Financial constraints	0.020***	-0.009***	-0.005***	-0.002***	-0.001
	(0.003)	(0.002)	(0.001)	(0.001)	(0.000)
Intensity of R&D	-0.007	0.029**	0.026***	0.004	-0.003
	(0.027)	(0.015)	(0.009)	(0.009)	(0.012)
Manufacturing firms	0.046***	0.024***	0.004***	-0.006***	-0.017***
	(0.003)	(0.002)	(0.001)	(0.001)	(0.002)
Wholesale and retail trade firms	0.037***	0.021***	0.008***	0.002	0.007***
	(0.003)	(0.002)	(0.001)	(0.001)	(0.002)
Constant	0.750***	0.773***	0.801***	0.818***	0.822***
	(0.008)	(0.004)	(0.003)	(0.004)	(0.005)
Pseudo R2	0.07	0.07	0.07	0.09	0.10
N	23,070	23,070	23,070	23,070	23,070
Panel B: Big Data Technologies					
Adoption of Big data analysis and usage	-0.002	0.001	-0.004**	-0.005***	-0.009***
	(0.004)	(0.002)	(0.002)	(0.002)	(0.003)
Intensity of ICT	0.016	0.021*	0.013**	0.026***	0.049***
	(0.014)	(0.013)	(0.006)	(0.009)	(0.014)

Table 4: Impact of digital technologies on distance to frontier firms based on labour productivity





Intensity of ICT x Adoption of Big Data	0.011	-0.004	0.018	0.008	0.002
	(0.076)	(0.034)	(0.024)	(0.016)	(0.076)
Pseudo R2	0.08	0.08	0.07	0.08	0.11
Ν	11,842	11,842	11,842	11,842	11,842
Panel C: Robotic Technologies					
Adoption of robotic technologies	0.011**	0.002	-0.004	-0.009***	-0.022***
	(0.005)	(0.003)	(0.003)	(0.003)	(0.005)
Intensity of ICT	0.015	0.019	0.013	0.010	0.025
	(0.024)	(0.019)	(0.008)	(0.009)	(0.015)
Intensity of ICT x Adoption of robots	-0.003	-0.035	-0.057	0.010	-0.020
	(0.092)	(0.053)	(0.097)	(0.103)	(0.112)
Pseudo R2	0.07	0.07	0.07	0.08	0.11
Ν	6,037	6,037	6,037	6,037	6,037

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*,

1%. All models included the control variables showed on Panel A.

Again, different digital technologies affect differently the dynamics of productivity at firm level, in particular the convergence to frontier. Specifically, there is some converge in laggard firms through input deepening in ERP and CRM technologies and in robotics. Laggard firms seem to benefit from the adoption of robotic technologies and ERM and CRM technologies in the initial (or, take-off) phase of the adoption process. Such adoption seems to have scope to mitigate the productivity gap to frontier firms, suggesting that they are able to re-start the engine of productivity growth. Nonetheless, this positive effect in the initial (or, take-off) phase is vanished as firms move up in the productivity distribution, indicating that persistent efforts to close the productivity gap require also investments in complementary resources such as inhouse ICT capabilities and more advanced ICT assets. In fact, firms at the end of the distribution - closer to and in the frontier -, are being able to attain even larger productivity gains than laggards through ERP and CRM technologies only when they complement it with internal ICT capabilities. Thus, these results suggest that convergence is weak. The absence of convergence in big data technologies could be due to the early stage diffusion.

#### **5.2 Robustness checks**

Here, we explore the robustness of our findings and disclose the possibility of heterogenous effects across different firms and industries. For that, several partitions of the data were made and the models re-estimated on different subsamples. In particular, we assess whether inter-firm and inter-industry heterogenous effects emerge, regarding (i) firm's size, (ii) technological intensity at industry level, and (iii) digital intensity at industry level. Further, the impact of digital technologies on productivity is re-assessed in order to evaluate whether productivity gains could be derived from interactions with other digitalised firms in the same industry.

#### 5.2.1 Inter-firm and inter-industry heterogeneity

The impact of digital technologies can be heterogeneous across different firm sizes and industries technology intensity. On one hand, it has been largely discussed that the use of a



single type of ICT asset may not be a strategic resource because competitors also have access to it (Walsh et al., 2010). So, combining various types of ICT and creating complementarity can be decisive for a firm (Chae et al., 2014). Likewise, the notion underlying the productivity J-curve (Brynjolfsson et al., 2021)- in early stages of new general purpose technologies firms experience a learning-by-doing process, implies that firms with more capabilities will benefit more from the adoption of digital technologies. On the other hand, recent evidence (Cataldo et al., 2020) has found firm's size moderates the impact of ICT on productivity: the smaller the company, the more significant the benefits of ICT assets.

We next explore the possibility of heterogeneous effects across different firm sizes. Table 5 shows the impact of digital technologies on productivity of SMEs and Table 6 present the estimates for the impact of digital technologies adoption to closing the distance to frontier firms among SMEs.

For SMEs, increases in firm productivity result mainly from the adoption of ERP and CRM technologies. These are the less technologically complex technologies and most widely spread. Nonetheless, the productivity impact of these technologies only happens when in interaction with the firms' complementary ICT assets investment, as shown by the interaction term. In line with previous contributions (Ballestar et al., 2020; Cataldo et al., 2020; Chun et al., 2015; Corrado et al., 2017; Pieri et al., 2018), we conclude that it is not the use of ERP and CRM which impacts on firms' productivity, but rather the degree of sophistication and combinations of use ICT-based capabilities (proxied by Intensity of ICT).





#### Table 5: Impact of digital technologies on labour productivity - SMEs firms

			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	-0.008	-0.006	-0.029	-	-0.038
	(0.030)	(0.016)	(0.018)	(0.017	) (0.025)
Intensity of ICT	1.190***	0.901	1.601***	1.698***	1.704***
	(0.313)	(0.625)	(0.610)	(0.628)	(0.605)
Intensity of ICT x Adoption of ERP and CRM tecnologies	2.416**	3.834***	4.339***	4.807***	4.486***
	(1.023)	(0.892)	(0.698)	(0.875)	(0.810)
Pseudo R2	0.16	0.21	0.24	0.28	0.34
Ν	11,614	11,614	11,614	11,614	11,614
Panel B: Big Data Technologies					
Adoption of Big data analysis and usage	-0.015	-0.039	-0.033*	-0.092***	-0.163***
	(0.034)	(0.024)	(0.019)	(0.026)	(0.045)
Intensity of ICT	4.516***	5.550***	7.022***	8.908***	9.212***
	(0.720)	(0.359)	(0.552)	(0.864)	(1.231)
Intensity of ICT x Adoption of Big Data	-4.113***	-5.084***	-5.226***	-5.274***	-4.343***
	(1.080)	(1.050)	(1.529)	(1.620)	(1.631)
Pseudo R2	0.15	0.20	0.23	0.27	0.32
Ν	6,482	6,482	6,482	6,482	6,482
Panel C: Robotic Technologies					
Adoption of robotic technologies	0.083*	0.035	-0.009	-0.104**	-0.185**
	(0.050)	(0.043)	(0.032)	(0.048)	(0.073)
Intensity of ICT	3.000***	4.072***	4.735***	6.456***	6.253**
	(1.017)	(1.135)	(1.308)	(1.536)	(2.533)
Intensity of ICT x Adoption of robots	1.003	-0.691	-1.951	-1.241	1.312
	(3.046)	(1.612)	(2.437)	(4.037)	(4.774)
Pseudo R2	0.14	0.19	0.22	0.26	0.31
Ν	3,197	3,197	3,197	3,197	3,197

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*, 1%. All models included the control variables *Firm's size*, *Export intensity*, *Financial constraints*, *Intensity of R&D* and sectoral dummies.

# Table 6: Impact of digital technologies on distance to frontier firms based on labour productivity - SMEs firms

			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	-0.002	-0.004**	-0.004**	-0.007***	-0.010***
	(0.003)	(0.002)	(0.002)	(0.002)	(0.002)
Intensity of ICT	-0.069	-0.064	-0.019	0.002	0.009
	(0.080)	(0.085)	(0.045)	(0.058)	(0.044)
Intensity of ICT x Adoption of ERP and CRM technologies	0.086	0.148*	0.133***	0.203***	0.369***
	(0.097)	(0.089)	(0.051)	(0.073)	(0.065)
Pseudo R2	0.09	0.07	0.07	0.09	0.12
Ν	11,614	11,614	11,614	11,614	11,614
Panel B: Big Data Technologies					
Adoption of Big data analysis and usage	-0.012***	0.000	-0.001	-0.002	-0.005
	(0.004)	(0.003)	(0.002)	(0.002)	(0.003)
Intensity of ICT	0.137**	0.158**	0.264***	0.387***	0.645***
	(0.069)	(0.065)	(0.074)	(0.079)	(0.119)
Intensity of ICT x Adoption of Biga Data	-0.112	-0.182**	-0.316***	-0.345***	-0.550***
	(0.075)	((0.072)	(0.102)	(0.131)	(0.165)
Pseudo R2	0.10	0.08	0.07	0.08	0.12
Ν	6,482	6,482	6,482	6,482	6,482
Panel C: Robotic Technologies					
Adoption of robotic technologies	0.005	-0.001	-0.004	-0.005	-0.009*
	(0.006)	(0.004)	(0.004)	(0.004)	(0.005)
Intensity of ICT	0.021	0.079	0.134*	0.246**	0.517***
	(0.058)	(0.073)	(0.081)	(0.104)	(0.200)





Intensity of ICT x Adoption of robots	0.088 (0.218)	-0.033 (0.204)	-0.171 (0.244)	-0.085 (0.199)	-0.480* (0.279)
Pseudo R2	0.10	0.07	0.06	0.08	0.13
Ν	3,197	3,197	3,197	3,197	3,197

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*, 1%. All models included the control variables *Firm's size*, *Export intensity*, *Financial constraints*, *Intensity of R&D* and sectoral dummies.

More importantly, perhaps, is the increasing impact of ERP and CRM technologies along firms' productivity distribution – up to the 75 quantile, with a slight decrease among frontier firms. This result strongly indicates that the benefits from the adoption of digital technologies increase with firm efficiency. Firms that already are more efficient in their production process, reap more benefits, which is in line with the argument that these technologies require internal capabilities (Bresnahan et al., 2002; Bloom et al., 2012). Regarding the other technologies, only robotics is having a positive impact on firm productivity. So far, this effect is limited to laggard firms and is independent from other complementary investments in ICT.

Table 6 shows the impact of digital technologies adoption in closing the productivity gap. Our findings indicate that the impact of digital technologies on productivity convergence to frontier among SMEs is limited to ERP and CRM technologies. However, results also show that convergence in productivity is increasing with firm's own productivity and that laggard firms are not being able to move into the frontier. Hence, these results indicate that there is taking place an aggregate movement towards the frontier but still with significant between-firm differences. More recent digital technologies seem to not be able to engender a convergence effect on productivity among SMEs, with a significant impact on the aggregate productivity as they represent a high proportion of firms across all sectors in the Portuguese economy.

Another reason why there are heterogeneous returns to digital technologies across firms is because they operate in very diverse industries. It has been documented that firms operating in high-tech sectors are able to reap larger benefits from more technologically advanced technologies such as ICT (Corrado et al., 2017; Stiroh, 2002) as they already operate in a context of technologically advanced technologies and hence are more technologically experienced than firms operating in low tech-sectors. Likewise, a similar effect could be observed in digital intensive industries, where firms are more experienced in using and combining to various types of ICT (Chae et al., 2014). Recent evidence (Pieri et al., 2018) has found a positive impact of ICT high-tech sectors and insignificant in low-tech industries.

Thus, we further explore differences across industries technological and digital intensity. Table 7 shows how digital adoption impacts on productivity of firms operating in high-tech sectors. Although limited, we still observe that input deepening in Big data technologies is contributing to increase firms productivity. Yet this effect is limited to firms with higher efficiency levels (situated in the 75 percentile of the productivity distribution). Also, in robotics can be observed a positive impact on productivity, but in the case of these technologies it only happens in interaction with other ICT investments, thereby corroborating the argument that these technologies need to be implemented with complementary assets.





			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	0.048	-0.020	-0.172***	-0.224***	-0.292***
	(0.089)	(0.052)	(0.059)	(0.074)	(0.107)
Intensity of ICT	0.622**	0.520***	0.440**	0.692***	0.362*
	(0.260)	(0.165)	(0.184)	(0.189)	(0.196)
Intensity of ICT x Adoption of ERP and CRM tecnologies	-0.367	-0.296	0.136	0.193	0.565
	(0.337)	(0.212)	(0.284)	(0.385)	(0.436)
Pseudo R2	0.20	0.22	0.22	0.24	0.27
Ν	1,462	1,462	1,462	1,462	1,462
Panel B: Big Data Technologies					
Adoption of Big data analysis and usage	0.132	0.105	0.090	0.259***	0.157
	(0.120)	(0.074)	(0.069)	(0.100)	(0.131)
Intensity of ICT	-0.060	0.504***	0.635***	0.793***	0.780***
	(0.582)	(0.167)	(0.115)	(0.182)	(0.175)
Intensity of ICT x Adoption of Big Data	-2.276	-3.017**	-0.959	-0.512	-0.830
	(1.400)	(1.311)	(0.962)	(1.251)	(1.499)
Pseudo R2	0.20	0.23	0.23	0.24	0.23
Ν	748	748	748	748	748
Panel C: Robotic Technologies					
Adoption of robotic technologies	-0.217	-0.354**	-0.382	-0.173	-0511**
	(0.328)	(0.169)	(0.218)	(0.317)	(0.249)
Intensity of ICT	0.209	0.447	0.525***	0.433**	0.339
	(0.680)	(0.326)	(0.126)	(0.210)	(0.229)
Intensity of ICT x Adoption of robots	5.001	2.995	7.622*	6.953	6.378
	(9.347)	(4.424)	(4.314)	(5.156)	(4.963)
Pseudo R2	0.18	0.23	0.24	0.24	0.26
Ν	398	398	398	398	398

#### Table 7: Impact of digital technologies on labour productivity - High-tech firms

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*, 1%. All models included the control variables *Firm's size*, *Export intensity*, *Financial constraints*, *Intensity of R&D* and sectoral dummies.

In Table 8 we investigate the impact of digital adoption in reducing productivity dispersion among firms in high-tech sectors. Again, we only observe statistically significant impacts among big data and robotics technologies. Big data investments are contributing to reduce the productivity gap but only among firms situated in the lower band of the productivity distribution - the 25 percentile. As for robotics, there is wider and significant impact, as firms with higher efficiency levels are clearly moving towards the frontier as they investment in robotics in complement with other ICT assets. Despite this positive outcome, it still is very restricted to a small part of the distribution of firms, which implies that at the aggregate level we do not observe a significant shift towards the frontier.





Table	8:	Impact	of	digital	technologies	on	distance	to	frontier	firms	based	on	labour
produc	ctivi	ity - High	ı-te	ch firms	5								

			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	0.007	-0.003	-0.010**	-0.015**	-0.020***
	(0.009)	(0.006)	(0.005)	(0.007)	(0.008)
Intensity of ICT	0.063**	0.048***	0.051***	0.068***	0.050***
	(0.028)	(0.016)	(0.016)	(0.019)	(0.013)
Intensity of ICT x Adoption of ERP and CRM tecnologies	-0.015	-0.016	-0.001	0.002	0.030
	(0.035)	(0.023)	(0.027)	(0.030)	(0.028)
Pseudo R2	0.16	0.16	0.17	0.15	0.18
Ν	1,462	1,462	1,462	1,462	1,462
Panel B: Big Data Technologies					
Adoption of Big data analysis and usage	0.015	0.013**	0.008	0.013	0.010
	(0.019)	(0.006)	(0.005)	(0.009)	(0.009)
Intensity of ICT	0.043	0.068***	0.056***	0.078***	0.070***
	(0.043)	(0.017)	(0.011)	(0.020)	(0.013)
Intensity of ICT x Adoption of Big Data	-0.220*	-0.224*	-0.083	-0.037	-0.069
	(0.124)	(0.118)	(0.093)	(0.090)	(0.065)
Pseudo R2	0.16	0.17	0.17	0.15	0.17
Ν	748	748	748	748	748
Panel C: Robotic Technologies					
Adoption of robotic technologies	-0.019	-0.029*	-0.026	-0.033	-0.052*
	(0.031)	(0.015)	(0.017)	(0.022)	(0.030)
Intensity of ICT	0.050	0.035	0.048***	0.028*	0.017
	(0.040)	(0.031)	(0.015)	(0.016)	(0.017)
Intensity of ICT x Adoption of robots	0.749	0.302	0.593	1.056*	0.976*
	(2.262)	(0.703)	(0.548)	(0.570)	(0.584)
Pseudo R2	0.14	0.17	0.17	0.17	0.21
Ν	398	398	398	398	398

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*, 1%. All models included the control variables *Firm's size*, *Export intensity*, *Financial constraints*, *Intensity of R&D* and

sectoral dummies.

Looking at firms operating in digital intensive industries, Tables 9 and 10 show the estimates of the impact o digital technologies adoption for those firms. Results in Table 9 show that only ERM and CRM technologies play a role in explaining firms differences in productivity. Specifically, investments in these technologies when complemented with other assets are leading to increases in productivity of almost all firms, but the benefits are also increasing with firms existing efficiency. The fact that we do not observe any impact of big data and robotics in firms operating in these industries can be explained by the fact that the users of big data and robotics.





#### Table 9: Impact of digital technologies on labour productivity - Digital intensive firms

			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	-0.084	-0.105***	-0.114***	-0.156***	-0.250***
	(0.053)	(0.036)	(0.024)	(0.034)	(0.057)
Intensity of ICT	0.236	0.423**	0.510***	0.694***	0.465**
	(0.223)	(0.165)	(0.122)	(0.168)	(0.205)
Intensity of ICT x Adoption of ERP and CRM tecnologies	0.497*	0.205	0.407*	0.402	1.222***
	(0.263)	(0.223)	(0.232)	(0.273)	(0.394)
Pseudo R2	0.08	0.09	0.14	0.18	0.21
Ν	4,371	4,371	4,371	4,371	4,371
Panel B: Big Data Technologies					
Adoption of Big data analysis and usage	0.016	0.029	0.007	-0.046	-0.168*
	(0.090)	(0.075)	(0.044)	(0.050)	(0.095)
Intensity of ICT	0.378	0.566***	0.684***	0.747***	0.780***
	(0.446)	(0.182)	(0.098)	(0.136)	(0.142)
Intensity of ICT x Adoption of Biga Data	-1.841	-0.656	-0.443	-0.239	-0.475
	(1.224)	(1.038)	(0.480)	(0.474)	(1.370)
Pseudo R2	0.07	0.09	0.15	0.18	0.19
N	2,225	2,225	2,225	2,225	2,225
Panel C: Robotic Technologies					
Adoption of robotic technologies	0.101	0.065	-0.027	0.019	-0.151
	(0.200)	(0.118)	(0.099)	(0.132)	(0.165)
Intensity of ICT	0.373	0.625**	0.576***	0.516***	0.298*
	(0.504)	(0.317)	(0.112)	(0.129)	(0.174)
Intensity of ICT x Adoption of robots	0.154	-0.930	-1.262	-0.400	-0.234
	(3.487)	(3.549)	(4.619)	(5.901)	(3.603)
Pseudo R2	0.07	0.09	0.17	0.20	0.20
Ν	1,118	1,118	1,118	1,118	1,118

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*, 1%. All models included the control variables *Firm's size, Export intensity, Financial constraints, Intensity of R&D* and sectoral dummies.





# Table 10: Impact of digital technologies on distance to frontier firms based on labour productivity - Digital intensive firms

			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	-0.010*	-0.011***	-0.011***	-0.009***	-0.013***
	(0.005)	(0.004)	(0.004)	(0.003)	(0.004)
Intensity of ICT	0.005	0.033*	0.010	0.031**	0.033**
	(0.017)	(0.018)	(0.011)	(0.015)	(0.015)
Intensity of ICT x Adoption of ERP and CRM technologies	0.033	0.014	0.018	0.011	0.031
	(0.026)	(0.024)	(0.023)	(0.025)	(0.032)
Pseudo R2	0.09	0.09	0.08	0.10	0.13
N	4,371	4,371	4,371	4,371	4,371
Panel B: Big Data Technologies					
Adoption of Big data analysis and usage	-0.004	0.005	-0.007	-0.003	-0.008
	(0.011)	(0.005)	(0.004)	(0.006)	(0.006)
Intensity of ICT	-0.011	0.035**	0.025***	0.033**	0.063***
	(0.032)	(0.016)	(0.009)	(0.013)	(0.014)
Intensity of ICT x Adoption of Biga Data	-0.109	-0.040	-0.010	0.009	-0.030
	(0.105)	(0.089)	(0.051)	(0.054)	(0.065)
Pseudo R2	0.08	0.09	0.09	0.09	0.12
Ν	2,225	2,225	2,225	2,225	2,225
Panel C: Robotic Technologies					
Adoption of robotic technologies	0.014	0.002	0.011	0.017	-0.001
	(0.011)	(0.008)	(0.010)	(0.012)	(0.011)
Intensity of ICT	0.002	0.017	0.022	0.006	0.014
	(0.034)	(0.026)	(0.016)	(0.012)	(0.017)
Intensity of ICT x Adoption of robots	0.067	0.006	-0.106	-0.130	-0.171
	(0.556)	(0.318)	(0.395)	(0.269)	(0.308)
Pseudo R2	0.09	0.10	0.10	0.11	0.14
Ν	1,118	1,118	1,118	1,118	1,118

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*, 1%. All models included the control variables *Firm's size*, *Export intensity*, *Financial constraints*, *Intensity of R&D* and sectoral dummies.

#### 5.2.2 Industry spillovers

An important channel through which firms can increase their productivity is by benefiting from industry spillovers, that is, benefits derived from interactions with digitalised peers in the same industry. For that, the models were re-estimated with the addition of an industry spillover variable, which is measured by the intensity of use of each digital technology at the industry in which the firm operates. Table 11 shows the estimated impact to firm's productivity from interactions with digitalised peers for all firms in the sample.





			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies	•				<i>.</i>
Adoption of ERP and CRM technologies	-0.054	-0.030	-0.090***	-0.155***	-0.095**
	(0.041)	(0.022)	(0.024)	(0.032)	(0.040)
Intensity of ICT	0.187	0.299**	0.394***	0.561***	0.458***
	(0.131)	(0.127)	(0.079)	(0.123)	(0.078)
Intensity of ICT x Adoption of ERP and CRM	0.288*	0.245	0.502***	0.973***	1.948***
	(0.158)	(0.152)	(0.152)	(0.254)	(0.423)
ERP and CRM at industry	0.361***	0.384***	0.425***	0.464***	0.526***
	(0.050)	(0.036)	(0.032)	(0.044)	(0.054)
ERP and CRM at industry x Adoption of ERP and CRM	0.013	-0.026	0.015	0.004	-0.293***
	(0.063)	(0.040)	(0.043)	(0.065)	(0.074)
Pseudo R2	0.16	0.18	0.21	0.24	0.27
N	23,070	23,070	23,070	23,070	23,070
Panel B: Big Data Technologies					
Adoption of big data analysis and usage	0.101**	0.003	-0.022	-0.124***	-0.207***
	(0.048)	(0.030)	(0.023)	(0.034)	(0.062)
Intensity of ICT	0.209	0.378***	0.624***	0.736***	0.899***
,	(0.184)	(0.103)	(0.095)	(0.106)	(0.167)
Intensity of ICT x Adoption of big data analysis and	0.257	0.433*	0.375*	1.637**	2.913***
	(0.501)	(0.246)	(0.214)	(0.764)	(0.664)
Big Data at industry	0.461***	0.401***	0.443***	0.609***	0.576***
,	(0.110)	(0.067)	(0.076)	(0.099)	(0.184)
Big Data at industry x Adoption of big data analysis and	-0.623**	-0.290**	-0.354***	-0.339**	-0.412
	(0.258)	(0.127)	(0.097)	(0.133)	(0.328)
Pseudo R2	0.15	0.18	0.21	0.23	0.27
N	11,842	11,842	11,842	11,842	11,842
Panel C: Robotic Technologies	11/012	11/012	11/012	11/0.12	11/012
Adoption of robotic technologies	-0.028	-0.030	-0.010	-0.165***	-0.302***
	(0.075)	(0.067)	(0.044)	(0.049)	(0.108)
Intensity of ICT	0.345	0.574***	0.775***	0.887***	1.138***
	(0.268)	(0.167)	(0.085)	(0.143)	(0.441)
Intensity of ICT x Adoption of robotic technologies	0.279	0.728	0.896	0.886	2.174
	(1.232)	(1.129)	(0.953)	(1.217)	(1.780)
Robots at industry	0.377***	0.306***	0.419***	0.552***	0.457**
	(0.137)	(0.113)	(0.114)	(0.147)	(0.227)
Robots at industry x Adoption of robotic technologies	0.117	-0.010	-0.340**	-0.368*	-0.369
Repeter at madely x Adoption of repetic technologies	(0.200)	(0.160)	(0.135)	(0.188)	(0.300)
Pseudo R2	0.14	0.17	0.21	0.23	0.26
N	6,037	6,037	6,037	6,037	6,037

#### Table 11: Impact of digital technologies on labour productivity - industry spillovers

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*, 5%; \*\*\*, 1%. All models included the control variables *Firm's size, Export intensity, Financial constraints, Intensity of R&D* and sectoral dummies.

Our findings suggest that there are significant industry spillovers across all digital technologies as well as across the productivity distribution. These benefits are also increasing with firms ' level of production efficiency, thereby corroborating previous findings in that more efficient firms tend to benefit more from digital adoption. Whereas this relationship is linear in the case of ERP and CRM technologies, in the case of big data and robotics there is a slightly larger impact among greater performers (firms in the 75 percentile of the productivity distribution), highlighting the fact that digital technologies are not homogenous entity and thereby their impact on firm productivity is also diverse. Nonetheless, as digital adoption at industry level increases, the lesser are productivity gains extracted from digital technologies adoption at firm level. This interesting finding suggests that adopters are not able to



appropriate all productivity benefits related to digital technologies as they spread out to all firms in the industry. It is also worth noting that laggard firms appear to benefit from big data analysis and usage when they are the first adopters in the industry, which could be interpreted as a first-mover advantage for laggard firms. As competitors also have access to digital technologies, the economic value of digital technologies seems to vanish.

A complementary point is to examine whether industry spillovers promote a productivity convergence effect, which would impact positively at aggregate productivity. In Table 12 estimates show in what extent industry spillovers are contributing to productivity convergence.

Table 12:	Impact	of	digital	technologies	on	distance	to	frontier	firms	based	on	labour
productivit	y - indus	try	spillove	ers								

· · · · · ·			Quantiles		
	10 (Laggard firms)	25	50	75	90 (Frontier firms)
Panel A: ERM and CRM Technologies					
Adoption of ERP and CRM technologies	0.034***	0.008**	-0.002	-0.008***	0.004
	(0.006)	(0.003)	(0.002)	(0.002)	(0.003)
Intensity of ICT	0.006	0.017	0.022***	0.025**	0.029***
	(0.011)	(0.012)	(0.006)	(0.010)	(0.008)
Intensity of ICT x Adoption of ERP and CRM technologies	0.008	0.013	0.020	0.025	0.060***
	(0.018)	(0.014)	(0.013)	(0.016)	(0.018)
ERP and CRM at industry	-0.023***	-0.056***	-0.062***	-0.058***	-0.043***
	(0.007)	(0.005)	(0.003)	(0.004)	(0.004)
ERP and CRM at industry x Adoption of ERP and CRM	-0.041***	-0.004	0.009**	0.015***	-0.012*
	(0.011)	(0.006)	(0.004)	(0.005)	(0.006)
Pseudo R2	0.06	0.08	0.09	0.10	0.12
Ν	23,070	23,070	23,070	23,070	23,070
Panel B: Big Data Technologies					
Adoption of big data analysis and usage	0.024***	0.004	-0.005	-0.004	-0.007
	(0.005)	(0.004)	(0.003)	(0.003)	(0.004)
Intensity of ICT	0.008	0.019*	0.018***	0.029***	0.042***
	(0.016)	(0.010)	(0.006)	(0.009)	(0.012)
Intensity of ICT x Adoption of big data analysis and	0.017	0.001	0.013	0.011	0.047
	(0.036)	(0.026)	(0.018)	(0.018)	(0.059)
Big Data at industry	-0.051***	-0.058***	-0.068***	-0.044***	-0.032**
5 1	(0.016)	(0.009)	(0.008)	(0.010)	(0.013)
Big Data at industry x Adoption of big data analysis and	-0.091***	-0.000	0.028**	0.016	-0.001
	(0.028)	(0.022)	(0.012)	(0.013)	(0.017)
Pseudo R2	0.07	0.08	0.08	0.09	0.11
Ν	11,482	11,482	11,482	11,482	11,482
Panel C: Robotic Technologies	, -	, -	, -	, -	, -
Adoption of robotic technologies	0.002	-0.007	-0.009*	-0.009	-0.013*
	(0.009)	(0.006)	(0.004)	(0.006)	(0.008)
Intensity of ICT	0.000	0.011	0.007	0.012	0.025
····, ···,	(0.018)	(0.015)	(0.010)	(0.010)	(0.016)
Intensity of ICT x Adoption of robotic technologies	0.033	-0.020	-0.050	0.004	-0.053
	(0.067)	(0.049)	(0.064)	(0.114)	(0.154)
Robots at industry	-0.039	-0.065***	-0.050***	-0.034**	-0.024
	(0.027)	(0.015)	(0.009)	(0.014)	(0.020)
Robots at industry x Adoption of robotic technologies	0.037	0.050**	0.038***	0.020	0.009
	(0.029)	(0.021)	(0.012)	(0.020)	(0.023)
Pseudo R2	0.05	0.06	0.07	0.08	0.11
N	6,037	6,037	6,037	6,037	6,037
Notes: Standard errors are computed by bootstrap based on 100	,			,	

Notes: Standard errors are computed by bootstrap based on 100 replications. Significance levels: \*, 10%; \*\*\*, 5%; \*\*\*\*, 1%. All models included the control variables Firm's size, Export intensity, Financial constraints, Intensity of R&D and sectoral dummies.





The first interesting result emerging from these estimates is that convergence only occurs through the interaction term between firm's own adoption and the industrys intensity use of the particular technology. So, as technologies diffuse across markets and industries the spillover effect from these stock effects (Colombo and Mosconi, 1995) will increase firm's productivity if the firm itself invests in these technologies.

As we observe some convergence effect across all types of digital technologies, their size and breath are different across technologies and firms. Now, there seems to be more convergence among ERP, CRM and robotics than among big data technologies. Also, firms situated in the median of the distribution, are benefiting more from these spillovers, than firms located at the ends of the productivity distribution.

Looking at laggard firms, the first-mover advantage seems to also emerge in terms of convergence to productivity frontier. When they are the first adopters in the industry of ERP and CRM technologies or big data, they appear to follow a pattern of convergence to frontier. This positive effect, however, is eliminated as more firms in the industry adopt this digital technology, suggesting again that an appropriation issue is at work and reinforcing the economic value of industry spillovers.

#### 6. Concluding remarks and policy recommendations

Digital technologies, in particular the latest generation of new technologies, which allow a faster adaptation to the changing productive and business environment and are adopted in different areas of the economy, are considered pathbreaking innovations with important expected effects on productivity and growth. Using a large-scale, representative sample of Portuguese firms, operating in manufacturing and services industries, between 2014 and 2019, we performed a quantile regression analysis relating productivity at firm-level to the adoption of different digital technologies in order to analyse the new sources of productivity and the scope of digital technologies to close the productivity gap between frontier and laggard firms.

Our main contribution is the uncovering of heterogenous productivity effects across different digital technologies and different types of firms and industries, which offer important insights in explaining the productivity aggregate effect of digital technologies. Overall, we found that heterogenous digital technologies affect differently the dynamics of productivity and the convergence to the frontier. The persistent efforts to close the productivity gap require an upgrading on the degree of sophistication and complementarities among digital technologies and other productivity-enhancing resources. The lack of such combination could shrink productivity, which seems to be frequent in the early phase of digital technology adoption and empirically corroborating the view of a J-curve patterns of the relationship between digital technologies and productivity.

More importantly, frontier firms appear to be the greater extractors of productivity gains associated to digital technologies, suggesting that digital technologies could be more likely to





exacerbate existing divides, rather than closing the productivity gap and improving aggregate productivity.

Looking at the impact of digital technologies on productivity among SMEs, which represent a high proportion of firms across all sectors in the Portuguese economy, new insights on the productivity convergence to frontier and its impact on aggregate productivity have emerged. Apart from ERP and CRM technologies, more mature technologies, second wave of digital technologies, such as big data and robots, seem yet to be not helpful in closing the productivity gap among SMEs. On the contrary, there is some evidence suggesting an increase in the existing divides, which would impact negatively in the aggregate productivity.

Moreover, the ability of firms to learn from more innovative firms at industry level seems to be at work and rendering productivity gains. Our findings indicate a visible effect on productivity associated to the rate at which digital technologies spread and replicate across industries. However, as that rate of adoption increases, firms seem to be less able to appropriate itself all the dynamics and benefits arising from digital technologies adoption. The diffusion of productivity benefits imply a less convergence rate but an expected positive effect on aggregate productivity.

As our findings were based on large-scale data that are representative for a wide variety of manufacturing and service industries, they are also valuable for policy makers and allow us to derive some policy implications. First, it is important to emphasise the need to coordinate efforts in the joint promotion of digital technologies adoptions and use with an upgrading in ICT use sophistication, organisational changes and human capital improvement. Digital technologies potential would not be realised without business and organisational changes and an increase in other complementary productivity-enhancing resources. Partial public policies to promote digitalisation at firm level without considering other investment in complementar resources and capabilities which affect productivity, may limit the expected positive effects at micro or aggregated level. Therefore, based on our findings, a more efficient choice and combination of policy initiatives and measures is desired.

Second, as digital technologies generate heterogenous productivity effects across firms of different size, public policies to promote digitalisation without considering this issue may not render positive impact at aggregate level, giving rise to the productivity paradox argument. For the Portuguese economy, it is a crucial point given the high proportion of SMEs across all sectors.

Third, apart from the positive impact on productivity associated to the rate of adoption at industry level, for laggard firms a first-move advantage related to digital technologies adoption is a finding that should deserve policy makers attention. In industries with no digital technologies adoption, policy initiatives and measures aiming to activate the technological diffusion machine are more prone to engender positive effects if they pay special attention to laggard firms, rather than to pick up frontier firms.





#### References

Acemoglu, D., Autor, D., Dorn, D., Hanson, G. H. and Price, B. (2014), 'Return of the solow paradox? IT, productivity, and employment in US manufacturing', *American Economic Review* 104(5), 394–99.

Akerman, A., Gaarder, I. and Mogstad, M. (2015), 'The skill complementarity of broadband internet', *The Quarterly Journal of Economics* 130(4), 1781–1824.

Andrews, D., Criscuolo, C. and Gal, P. (2015), 'Frontier firms, technology diffusion and public policy: Micro evidence from oecd countries', *OECD Productivity Working Paper* 2.

Andrews, D., Criscuolo, C. and Gal, P. (2016), 'The best versus the rest: The global productivity slowdown, divergence across firms and the role of public policy', *OECD Productivity Working Papers* 5.

Audretsch, D. B. and Belitski, M. (2020), 'The role of R&D and knowledge spillovers in innovation and productivity', *European Economic Review* 123, 103391.

Ballestar, M. T., D´ıaz-Chao, A., Sainz, J. and Torrent-Sellens, J. (2020), 'Knowledge,' robots and productivity in SMEs: Explaining the second digital wave', *Journal of Business Research* 108, 119–131.

Bartel, A., Ichniowski, C. and Shaw, K. (2007), 'How Does Information Technology Affect Productivity? Plant-Level Comparisons of Product Innovation, Process Improvement, and Worker Skills\*', *The Quarterly Journal of Economics* 122(4), 1721–1758.

Bartelsman, E. (2013), 'ICT, reallocation and productivity', *Economic Papers, European Commission* 486.

Bartelsman, E. J. and Doms, M. (2000), 'Understanding productivity: Lessons from longitudinal microdata', *Journal of Economic Literature* 38(3), 569–594.

Bartelsman, E., van Leeuwen, G. and Polder, M. (2017), 'CDM using a cross-country micro moments database', *Economics of Innovation and New Technology* 26(1-2), 168–182.

Berlingieri, G., Calligaris, S., Criscuolo, C. and Verlhac, R. (2020), 'Laggard firms, technology diffusion and its structural and policy determinants', *OECD Science, Technology and Industry Policy Papers* 86.

Bloom, N., Sadun, R. and Van Reenen, J. (2012), 'Americans do it better: US multinationals and the productivity miracle', *American Economic Review* 102(1), 167–201.

Bresnahan, T. F., ErikBrynjolfsson and Hitt, L. M. (2002), 'Information technology, workplace organization, and the demand for skilled labor: Firm-level evidence', *The Quarterly Journal of Economics* 117(1), 339–376.





Bresnahan, T. F. and Trajtenberg, M. (1995), 'General purpose technologies 'engines of growth'?', *Journal of Econometrics* 65(1), 83–108.

Brynjolfsson, E. and McAfee, A. (2012), *Race against the Machine: How the Digital Revolution Is Accelerating Innovation, Driving Productivity, and Irreversibly Transforming Employment and the Economy*, Digital Frontier Press.

Brynjolfsson, E. and McAfee, A. (2014), *The second machine age: Work, progress, and prosperity in a time of brilliant technologies*, W W Norton Co.

Brynjolfsson, E., Rock, D. and Syverson, C. (2021), 'The productivity J-curve: How intangibles complement general purpose technologies', *American Economic Journal: Macroeconomics* 13(1), 333–72.

Calvino, F., Criscuolo, C., Marcolin, L. and Squicciarini, M. (2018), 'A taxonomy of digital intensive sectors', *OECD Science, Technology and Industry Working Papers* No. 2018/14.

Cataldo, A., Pino, G. and McQueen, R. J. (2020), 'Size matters: the impact of combinations of ICT assets on the performance of chilean micro, small and medium enterprises', *Information Technology for Development* 26(2), 292–315.

Chae, H.-C., Koh, C. E. and Prybutok, V. R. (2014), 'Information technology capability and firm performance: Contradictory findings and their possible causes', *MIS Quarterly* 38(1), 305–325.

Chun, H., Kim, J.-W. and Lee, J. (2015), 'How does information technology improve aggregate productivity? a new channel of productivity dispersion and reallocation', *Research Policy* 44(5), 999–1016.

Colombo, M. G. and Mosconi, R. (1995), 'Complementarity and cumulative learning effects in the early diffusion of multiple technologies', *Journal of Industrial Economics* 43(1), 13–48.

Comin, D. and Mestieri, M. (2013), 'Technology diffusion: Measurement, causes and consequences', *NBER Working Papers* 19052.

Corrado, C., Haskel, J. and Jona-Lasinio, C. (2017), 'Knowledge spillovers, ICT and productivity growth', *Oxford Bulletin of Economics and Statistics* 79(4), 592–618.

Criscuolo, C., Gal, P. N. and Menon, C. (2014), 'The dynamics of employment growth:

New evidence from 18 countries', *OECD Science, Technology and Industry Policy Papers* 14.

DeStefano, T., Kneller, R. and Timmis, J. (2018), 'Broadband infrastructure, ICT use and firm performance: Evidence for uk firms', *Journal of Economic Behavior Organization* 155, 110–139.





Frank, A. G., Dalenogare, L. S. and Ayala, N. F. (2019), 'Industry 4.0 technologies: Implementation patterns in manufacturing companies', *International Journal of Production Economics* 210, 15–26.

Gabaix, X. and Landier, A. (2008), 'Why has CEO pay increased so much?', *The Quarterly Journal of Economics* 123(1), 49–100.

Gal, P., Nicoletti, G., Renault, T., Sorbe, S. and Timiliotis, C. (2019), 'Digitalisation and productivity: In search of the holy grail – firm-level empirical evidence from EU countries', *OECD Economics Department Working Papers* 1533.

Galindo-Rueda, F. and Verger, F. (2016), 'OECD taxonomy of economic activities based on R&D intensity', OECD Science, Technology and Industry Working Papers 2016/04.

Hall, B., Lotti, F. and Mairesse, J. (2009), 'Innovation and productivity in SMEs: empirical evidence for italy', *Small Business Economics* 33(1), 13–33.

Jovanovic, B. and Rousseau, P. (2005), General purpose technologies, *in* P. Aghion and S. Durlauf, eds, 'Handbook of Economic Growth', 1 edn, Vol. 1 of *1*, Elsevier, chapter 18, pp. 1181–1224.

Jungmittag, A. (2021), 'Robotisation of the manufacturing industries in the EU: Convergence or divergence?', *The Journal of Technology Transfer* 46(5), 1269–1290.

Koenker, R. (2005), *Quantile Regression*, Econometric Society Monographs, Cambridge University Press.

Malerba, F. and Orsenigo, L. (1996), 'Schumpeterian patterns of innovation are technologyspecific', *Research Policy* 25(3), 451–478.

Miyazaki, S., Idota, H. and Miyoshi, H. (2012), 'Corporate productivity and the stages of ICT development', *Information Technology and Management* 13, 17–26.

Mohnen, P. and Hall, B. H. (2013), 'Innovation and productivity: An update', *Eurasian Business Review* 3(1), 47–65.

Niebel, T., Rasel, F. and Viete, S. (2019), 'BIG data – BIG gains? understanding the link between big data analytics and innovation', *Economics of Innovation and New Technology* 28(3), 296–316.

Pieri, F., Vecchi, M. and Venturini, F. (2018), 'Modelling the joint impact of R&D and ICT on productivity: A frontier analysis approach', *Research Policy* 47(9), 1842–1852.

Pinheiro Alves, R. (2017), 'Portugal: A parodox in productivity', *International Productivity Monitor* 32, 137–160.





Stiroh, K. J. (2002), 'Information technology and the U.S. productivity revival: What do the industry data say?', *American Economic Review* 92(5), 1559–1576.

Syverson, C. (2011), 'What determines productivity?', *Journal of Economic Literature* 49(2), 326–65.

Syverson, C. (2017), 'Challenges to mismeasurement explanations for the US productivity slowdown', *Journal of Economic Perspectives* 31(2), 165–86.

Venturini, F. (2021), Intelligent technologies and productivity spillovers: Evidence from the fourth industrial revolution. University of Perugia.





## **Appendix A: Variables**

Table A.1: Variables,	measurement and source
-----------------------	------------------------

Variables	Measurement	Source
Dependent variables		
Labour Productivity	log of value added per worker	SCIE
Turnover	log of firm's turnover	SCIE
Distance to productivity frontier	Ratio of firm's labour productivity to 95th percentile of labour productivity distribution at industry-year level	SCIE
Digital technologies		
Adoption of ERM and CRM technologies	Indicator variable if the firm reports using enterprise resource planning (ERP) or customer relationship management (CRM)	IUTIC
Adoption of Big Data analysis and usage	Indicator variable if the firm report analysing and using big data	IUTIC
Adoption of robotic technologies	Indicator variable if the firm report working with robots	IUTIC
ICT intensity	Share of employees involved only in ICT activities	IUTIC
ERP and CRM at industry	Share of firms that have adopted ERP and/or CRM technologies at industry-year level	IUTIC
Big Data at industry	Share of firms that have adopted Big Data technologies at industry-year level	IUTIC
Robots at industry	Share of firms that have adopted robotic technologies at industry-year level	IUTIC
Control variables		
Firm's size	log of total assets	SCIE
Exporting intensity	Share of sales in international markets	SCIE
Financial constraints	Share of liabilities on total assets	SCIE
R&D intensity	Share of employees involved in R&D activities	SCIE
Manufacturing firms	Indicator variable if the firm belongs to NACE 10 to 33 industry	SCIE
Wholesale and retail trade firms	Indicator variable if the firm belongs to NACE 45 to 47 industry	SCIE





#### **GEE Papers**

- 1: Evolução do Comércio Externo Português de Exportação (1995-2004) João Ferreira do Amaral
- 2: Nowcasting an Economic Aggregate with Disaggregate Dynamic Factors: An Application to Portuguese GDP Antonio Morgado | Luis Nunes | Susana Salvado
- 3: Are the Dynamics of Knowledge-Based Industries Any Different? Ricardo Mamede | Daniel Mota | Manuel Godinho
- 4: Competitiveness and convergence in Portugal Jorge Braga de Macedo
- 5: Produtividade, Competitividade e Quotas de Exportação Jorge Santos
- 6: Export Diversification and Technological Improvement: Recent Trends in the Portuguese Economy Manuel Cabral
- 7: Election Results and Opportunistic Policies: An Integrated Approach Toke Aidt | Francisco Veiga | Linda Veiga
- 8: Behavioural Determinants of Foreign Direct Investment Ricardo Pinheiro-Alves
- 9: Structural Transformation and the role of Foreign Direct Investment in Portugal: a descriptive analysis for the period 1990-2005 Miguel de Freitas | Ricardo Mamede
- 10: Productive experience and specialization opportunities for Portugal: an empirical assessment Miguel de Freitas | Susana Salvado | Luis Nunes | Rui Costa Neves
- 11: The Portuguese Active Labour Market Policy during the period 1998-2003 - A Comprehensive Conditional Difference-In-Differences Application Alcina Nunes | Paulino Teixeira
- 12: Fiscal Policy in a Monetary Union: Gains from Changing Institutions Susana Salvado
- 13: Coordination and Stabilization Gains of Fiscal Policy in a Monetary Union Susana Salvado
- 14: The Relevance of Productive Experience in the Process of Economic Growth: an Empirical Study Diana Vieira
- 15: Employment and Exchange rates: the Role of Openness and Technology

Fernando Alexandre | Pedro Bação | João Cerejeira | Miguel Portela

- 16: Aggregate and sector-specific exchange rate indexes for the Portuguese economy Fernando Alexandre | Pedro Bação | João Cerejeira | Miguel Portela
- 17: The Macroeconomic Determinants of Cross Border Mergers and Acquisitions and Greenfield Investments Paula Neto | Antonio Brandao | António Cerqueira
- 18: Does the location of manufacturing determine service sectors' location choices? Evidence from Portugal Nuno Crespo | Maria Paula Fontoura
- 19: A hipótese do Investment Development Path: Uma Abordagem por Dados em Painel. Os casos de Portugal e Espanha Miguel Fonseca | António Mendonça | José Passos
- 20: Outward FDI Effects on the Portuguese Trade Balance, 1996-2007 Miguel Fonseca | António Mendonça | José Passos
- 21: Sectoral and regional impacts of the European Carbon Market in Portugal Margarita Robaina Alves | Miguel Rodriguez | Catarina Roseta-Palma
- 22: Business Demography Dynamics in Portugal: A Non-Parametric Survival Analysis Alcina Nunes | Elsa Sarmento
- 23: Business Demography Dynamics in Portugal: A Semi-parametric Survival Analysis Alcina Nunes | Elsa Sarmento
- 24: Digging Out the PPP Hypothesis: an Integrated Empirical Coverage Miguel de Carvalho | Paulo Júlio
- 25: Regulação de Mercados por Licenciamento Patrícia Cerqueira | Ricardo Pinheiro Alves
- 26: Which Portuguese Manufacturing Firms Learn by Exporting? Armando Silva | Óscar Afonso | Ana Paula Africano
- 27: Building Bridges: Heterogeneous Jurisdictions, Endogenous Spillovers, and the Benefits of Decentralization Paulo Júlio | Susana Peralta
- 28: Análise comparativa de sobrevivência empresarial: o caso da região Norte de Portugal Elsa Sarmento | Alcina Nunes





29: Business creation in Portugal: Comparison between the World Bank data and Quadros de Pessoal Elsa Sarmento | Alcina Nunes

 30: The Ease of Doing Business Index as a tool for Investment location decisions
 João Zambujal Oliveira | Ricardo Pinheiro Alves

- 31: The Politics of Growth: Can Lobbying Raise Growth and Welfare? Paulo Júlio
- 32: The choice of transport technology in the presence of exports and FDI José Pedro Ponte | Armando Garcia Pires
- 33: Tax Competition in an Expanding European Union

Ronald Davies | Johannes Voget

- 34: The usefulness of State trade missions for the internationalization of firms: an econometric analysis Ana Paula Africano | Aurora Teixeira | André Caiado
- 35: The role of subsidies for exports: Evidence from Portuguese manufacturing firms Armando Silva
- 36: Criação de empresas em Portugal e Espanha: análise comparativa com base nos dados do Banco Mundial Elsa Sarmento | Alcina Nunes
- 37: Economic performance and international trade engagement: the case of Portuguese manufacturing firms Armando Silva | Oscar Afonso | Ana Paula Africano
- 38: The importance of Intermediaries organizations in international R&D cooperation: an empirical multivariate study across Europe Aurora Teixeira | Margarida Catarino
- 39: Financial constraints, exports and monetary integration - Financial constraints and exports: An analysis of Portuguese firms during the European monetary integration Filipe Silva | Carlos Carreira
- 40: FDI and institutional reform in Portugal Paulo Júlio | Ricardo Pinheiro-Alves | José Tavares
- 41: Evaluating the forecast quality of GDP components
  Paulo Júlio | Pedro Esperança | João C. Fonseca
- 42: Assessing the Endogeneity of OCA conditions in EMU Carlos Vieira | Isabel Vieira

43: Labor Adjustment Dynamics: An Application of System GMM Pedro Esperança

- 44: Corporate taxes and the location of FDI in Europe using firm-level data Tomás Silva | Sergio Lagoa
- 45: Public Debt Stabilization: Redistributive Delays versus Preemptive Anticipations Paulo Júlio
- 46: Organizational Characteristics and Performance of Export Promotion Agencies: Portugal and Ireland compared Inês Ferreira | Aurora Teixeira
- 47: Evaluating the forecast quality of GDP components: An application to G7 Paulo Júlio | Pedro Esperança
- 48: The influence of Doing Business' institutional variables in Foreign Direct Investment Andreia Olival
- 49: Regional and Sectoral Foreign Direct Investment in Portugal since Joining the EU: A Dynamic Portrait Irina Melo | Alexandra Lopes
- 50: Institutions and Firm Formation: an Empirical Analysis of Portuguese Municipalities Simão Arouca
- 51: Youth Unemployment in Southern Europe João Leão | Guida Nogueira
- 52: Financiamento da Economia Portuguesa: um Obstáculo ao Crescimento? João Leão | Ana Martins | João Gonçalves
- 53: O Acordo de Parceria Transatlântica entre a UE e os EUA constitui uma ameaça ou uma oportunidade para a Economia Portuguesa? João Leão | Guida Nogueira
- 54: Prescription Patterns of Pharmaceuticals Ana Gonçalves
- 55: Economic Growth and the High Skilled: the Role of Scale Eects and of Barriers to Entry into the High Tech Pedro Gil | Oscar Afonso | Paulo Brito
- 56: Finanças Públicas Portuguesas Sustentáveis no Estado Novo (1933-1974)? Ricardo Ferraz
- 57: What Determines Firm-level Export Capacity? Evidence from Portuguese firms Ana Gouveia | Ana Luisa Correia
- 58: The effect of developing countries' competition on regional labour markets in Portugal Tiago Pereira
- 59: Fiscal Multipliers in the 21st century Pedro Brinca | Hans Holter | Per Krusell | Laurence Malafry
- 60: Reallocation of Resources between Tradable and Non-Tradable Sectors in Portugal: Developing a new Identification Strategy for the Tradable Sector Ana Fontoura Gouveia | Filipa Canas





- 61: Is the ECB unconventional monetary policy effective? Inês Pereira
- 62: The Determinants of TFP Growth in the Portuguese Manufacturing Sector Daniel Gonçalves | Ana Martins
- 63: Practical contribution for the assessment and monitoring of product market competition in the Portuguese Economy – estimation of price cost margins Luis Folque
- 64: The impact of structural reforms of the judicial system: a survey

Ana Gouveia | Silvia Santos | Corinna Herber

- 65: The short-term impact of structural reforms on productivity growth: beyond direct effects Ana Gouveia | Silvia Santos | Inês Gonçalves
- 66: Assessing the Competitiveness of the Portuguese Footwear Sector Fábio Batista | José Matos | Miguel Matos
- 67: The empirics of agglomeration economies: the link with productivity Ana Gouveia | Silvia Santos | Marli Fernandes
- 68: Determinants of the Portuguese GDP stagnation during the 2001-2014 period: an empirical investigation Carlos Figueira
- 69: Short-run effects of product markets' deregulation: a more productive, more efficient and more resilient economy? Ana Gouveia | Silvia Santos | Gustavo Monteiro
- 70: Portugal: a Paradox in Productivity Ricardo Pinheiro Alves
- 71: Infrastructure Investment, Labor Productivity, and International Competitiveness: The Case of Portugal Alfredo Pereira | Rui Pereira
- 72: Boom, Slump, Sudden stops, Recovery, and Policy Options. Portugal and the Euro Olivier Blanchard | Pedro Portugal
- 73: Case Study: DBRS Sovereign Rating of Portugal. Analysis of Rating Methodology and Rating Decisions Annika Luisa Hofmann | Miguel Ferreira | João Lampreia
- 74: For Whom the Bell Tolls: Road Safety Effects of Tolls on Uncongested SCUT Highways in Portugal Alfredo Pereira | Rui Pereira | João Pereira dos Santos
- 75: Is All Infrastructure Investment Created Equal? The Case of Portugal Alfredo Pereira | Rui Pereira
- 76: Why Virtuous Supply-Side Effects and Irrelevant Keynesian Effects are not Foregone

Conclusions: What we Learn from an Industry-Level Analysis of Infrastructure Investments in Portugal Alfredo Pereira | Rui Pereira

- 77: The Role of Gravity Models in Estimating the Economic Impact of Brexit Graham Gudgin | Ken Coutts | Neil Gibson | Jordan Buchanan
- 78: Infrastructure Investment in Portugal and the Traded/Non-Traded Industry Mix Alfredo Pereira | Rui Pereira
- 79: Goods and Factor Market Integration: A Quantitative Assessment of the EU Enlargement Lorenzo Caliendo | Fernando Parro | Luca David Opromolla | Alessandro Sforza
- 80: Understanding productivity dynamics:a task taxonomy approach Tiago Fonseca | Francisco Lima | Sonia C. Pereira
- 81: On the Effects of Infrastructure Investments on Industrial CO2 Emissions in Portugal Alfredo Pereira | Rui Pereira
- 82: Assessing Competition With the Panzar-Rosse Model: An empirical analysis of European Union banking industry Suzana Cristina Silva Andrade
- 83: Health Care Investments and Economic Performance in Portugal: An Industry Level Analysis Alfredo Pereira | Rui Pereira | Pedro G. Rodrigues
- 84: Is deregulation of product and labour markets promoting employment and productivity? A difference-in-differences approach Hugo Correia | Ana Fontoura Gouveia
- 85: Foreign acquisition and internal organization Paulo Bastos | Natália P. Monteiro | Odd Rune Straume
- 86: Learning, Prices, and Firm Dynamics Paulo Bastos | Daniel A. Dias | Olga A. Timoshenko
- 87: The Diffusion of Knowledge via Managers' Mobility Giordano Mion | Luca David Opromolla | Alessandro Sforza
- 88: Empresas Zombie em Portugal Os sectores não transacionáveis da Construção e dos Serviços
   Gabriel Osório de Barros | Filipe Bento Caires | Dora Xarepe Pereira
- 89: Collective bargaining through the magnifying glass: A comparison between the Netherlands and Portugal Alexander Hijzen | Pedro Martins | Jante Parlevliet





- 90: A Lower VAT Rate on Electricity in Portugal: Towards a Cleaner Environment, Better Economic Performance, and Less Inequality Alfredo Pereira | Rui Manuel Pereira
- 91: Who Seeks Re-Election: Local Fiscal Restraints and Political Selection Susana Peralta | João Pereira dos Santos
- 92: Assessing the Competitiveness of the Metalworking Sector João Marinho | Pedro Carvalho
- 93: The efficiency of Portuguese Technology Transfer Offices and the importance of university characteristics Aurora Teixeira | André Monteiro
- 94: Persistence in innovation and innovative behavior in unstable environments Joana Costa | Anabela Botelho | Aurora Teixeira
- 95: The effect of entrepreneurial origin on firms' performance - The case of Portuguese academic spinoffs Natália Barbosa | Ana Paula Faria
- 96: Absorptive Capacity and Firms' Generation of Innovation - Revisiting Zahra and George's Model Dina Pereira | João Leitão
- 97: Innovations in digital government as business facilitators: implications for Portugal João Martins | Linda Veiga
- 98: Innovation and the economic downturn: Insights from Portuguese firms Hugo Pinto | Tiago Santos Pereira | Elvira Uyarra
- 99: European Funds and Firm Dynamics: Estimating Spillovers from Increased Access João Pereira dos Santos | José Tavares
- 100: Corporate Leverage and Investment in Portugal Ana Martins | José Henrique Gonçalves | João Mário Ferreira Duque
- 101: The effects of official and unofficial information on tax compliance Filomena Garcia | Luca David Opromolla | Andrea Vezzulli | Rafael Marques
- 102: Competition effect on innovation and productivity - The Portuguese case Anabela Santos | Michele Cincera | Paulo Neto | Maria Manuel Serrano
- 103: Measuring the Welfare of Intermediation in Vertical Markets Javier D. Donna | Pedro Pereira | Tiago Pires | Andre Trindade
- 104: Of course Collusion Should be Prosecuted. But Maybe... Or (The case for international antitrust agreements) Filomena Garcia | Jose Manuel Paz y Minõ | Gustavo Torrens

- 105: Product market competition and gender discrimination Dudley Cooke | Ana P. Fernandes | Priscila Ferreira
- 106: Integration of Small Technology-Based Firms in Aeronautics Anabela Reis | Joana Mendonca | Ligia Urbina
- 107: The Effects of Highway Tolls on Private Business Activity – Results from a Natural Experiment João Pereira dos Santos | David B. Audretsch | Dirk Dohse
- 108: Competition and Firm Productivity: Evidence from Portugal Pedro Carvalho
- 109: Do Exchange Traded Funds (ETFs) Outperform the Market? Evidence from the Portuguese Stock Index Carlos Manuel Pinheiro | Hugo Hilário Varela
- 110: Assessing the Competitiveness of the Portuguese Chemical Sector Ana Rita Marques | Cátia Silva
- 111: A General Equilibrium Theory of Occupational Choice under Optimistic Beliefs about Entrepreneurial Ability Michele Dell'Era | Luca David Opromolla | Luis Santos-Pinto
- 112: O Mercado Segurador em Portugal: O Papel dos Gestores na Constituição de Provisões Soraia de Sousa Bornett | Carlos Manuel Pinheiro
- 113: Exploring the implications of di erent loan-tovalue macroprudential policy designs Rita Basto | Sandra Gomes | Diana Lima
- 114: The Determinants of TFP Growth in the Portuguese Service Sector Ana Martins | Tiago Domingues | Catarina Branco
- 115: Agglomeration and Industry Spillover Effects in the Aftermath of a Credit Shock José Jorge | Joana Rocha
- 116: Entrepreneurial Human Capital and Firm Dynamics Francisco Queiró
- 117: Global Value Chains and Vertical Specialization: The case of Portuguese Textiles and Shoes exports Tiago Domingues
- 118: Firm heterogeneity and exports in Portugal: Identifying export potential Frederico Oliveira Torres
- 119: Vantagens Comparativas Reveladas e suas determinantes: Uma Aplicação à Economia Portuguesa Guida Nogueira | António Portugal Duarte
- 120: A Look at the main channels of Potential Impact of Brexit on the Portuguese Economy





#### Guida Nogueira | Paulo Inácio

- 121: How internationalization and competitiveness contribute to get public support to innovation? The Portuguese case Anabela Santos, Michele Cincera, Paulo Neto | Maria Manuel Serrano
- 122: Grande Guerra e Guerra Colonial: Quanto Custaram aos Cofres Portugueses? Ricardo Ferraz
- 123: Financing a Renewable Energy Feed-in Tariff with a Tax on Carbon Dioxide Emissions: A Dynamic Multi-Sector General Equilibrium Analysis for Portugal Rui M. Pereira | Alfredo M. Pereira
- 124: Brown Sugar, how come you taste so good? The impact of a soda tax on prices and consumption Judite Gonçalves | João Pereira dos Santos
- 125: ARFIMA Reference Forecasts for Worldwide CO2 Emissions and the National Dimension of the Policy Efforts to Meet IPCC Targets José Beirute | Alfredo M. Pereira
- 126: Reference Forecasts for CO2 Emissions from Fossil-Fuel Combustion and Cement Production in Portugal José M. Belbute | Alfredo M. Pereira
- 127: Regulated Early Closures of Coal-Fired Power Plants and Tougher Energy Taxation on Electricity Production: Synergy or Rivalry? Alfredo Marvão Pereira | Rui Manuel Pereira
- 128: Picking Our Environmental Battles: Removal of Harmful Subsidies or Carbon Taxation? Alfredo Marvão Pereira | Rui Marvão Pereira
- 129: Financing Future Feed-in Tariffs from Currently Installed RES-E Generating Capacity Alfredo Marvão Pereira | Rui Marvão Pereira
- 130: Foreign Direct Investment, Income Inequality and Poverty in Portugal, 1973-2014: What does cointegration analysis tell us? Aurora Teixeira | Ana Sofia Loureiro
- 131: On the Spillover Effects of CO2 Taxation on the Emissions of other Air Pollutants Alfredo Marvão Pereira | Rui Marvão Pereira
- 132: On the Macroeconomic and Distributional Effects of the Regulated Closure of Coal-Operated Power Plants Alfredo Marvão Pereira | Rui Manuel Pereira
- 133: The China Shock and Employment in Portuguese Firms Lee Branstetter | Brian Kovak | Jacqueline Mauro | Ana Venâncio
- 134: Energy Taxation Reform with an Environmental Focus Alfredo Marvão Pereira | Rui Manuel Pereira
- 135: ARFIMA Reference Forecasts for Worldwide CO2 Emissions and the Need for Large and Frontloaded Decarbonization Policies

#### José M. Belbute | Alfredo M. Pereira

- 136: Exporter Firms Behaviour, Evidence From Portuguese Firms Using Microdata Luís Pedro Manso Machado
- 137: Collateral Value and Entrepreneurship: Evidence from a Property Tax Reform Miguel Ferreira | João Pereira dos Santos | Ana Venâncio
- 138: The Financial Channels of Labor Rigidities: Evidence from Portugal Edoardo M. Acabbi | Ettore Panetti | Alessandro Sforza
- 139: Can a small leak sink a great ship? A comprehensive analysis of the Portuguese household savings Tiago Domingues | Margarida Castro Rego
- 140: Corporate taxes and high-quality entrepreneurship: evidence from a tax reform Ana Venâncio | Victor Barros | Clara Raposo
- 141: Built Like a House of Cards? Corporate Indebtedness and Productivity Growth in the Portuguese Construction Sector1 José Santos | Nuno Tavares | Gabriel Osório de Barros
- 142: Effectiveness of Simplex: The Case of Portuguese Social Security António Alberto Nifrário de Pinho Tavares
- 143: Digital innovation in higher education: A questionnaire to Portuguese universities and polytechnic institutes Paulo Nuno Vicente |Margarida Lucas | Vânia Carlos
- 144: Portugal in the Global Innovation Index: A panel data analysis Marcelo P. Duarte | Fernando M. P. O. Carvalho
- 145: Intangible investments and productivity performance Michele Cincera | Julie Delanote | Pierre Mohnen | Anabela Santos | Christoph Weiss
- 146: Digitalization in Two-sided Platform Competition Filomena Garcia | Muxin Li
- 147: Collusion between two-sided platforms Joana Pinho | Yassine Lefouili
- 148: Da confluência entre Big Data e Direito da Concorrência: As concentrações digitais - O caso Facebook/WhatsApp Ana Rodrigues Bidarra
- 149: The Determinants of Total Factor Productivity in the Portuguese Quaternary Sector Paulo Matos | Pedro Neves
- 150: Os modelos Input-Output, a estrutura setorial das economias e o impacto da crise da COVID 19 Pedro N. Ramos | João Ferreira | Luís Cruz | Eduardo Barata





- 151: Public Expenditure and private firm performance: using religious denominations for causal inference Henrique Alpalhão | Marta Lopes | João Santos| José Tavares
- 152: Employee Training and Firm Performance: Quasi-experimental evidence from the European Social Fund Pedro S. Martins
- 153: Dream Jobs Luca David Opromolla | Giordano Mion | Gianmarco I.P. Ottaviano
- 154: Minimum wage and financially distressed firms: another one bites the dust F. Alexandre | P. Bação | J. Cerejeira | H. Costa | M. Portela
- 155: Do short-term rentals increase housing prices? Quasi-experimental evidence from Lisbon Susana Peralta | João Pereira dos Santos | Duarte Gonçalves
- 156: Economic and social policies under EMU Ricardo Pinheiro Alves

- 157: International Sourcing in Portuguese Companies - Evidence from Portuguese Micro Data Ana Martins | Guida Nogueira | Eva Pereira
- 158: The Impact of R&D tax incentives in Portugal Rita Bessone Basto | Ana Martins | Guida Nogueira
- 159: The Determinants of Competitiveness of the Portuguese Defense Industry Roxanne Merenda
- 160: How is the minimum waste shaping the wage distribution; bite, spillovers, and wage inequality Carlos Oliveira
- 161: Macroeconomy Impacts of the Covid-19 Pandemic in Some European Union Countries: a Counterfactual Analysis António Portugal Duarte | Fátima Sol Murta
- 162: Digital adoption and productivity: understanding micro drivers of the aggregate effect Natália Barbosa | Ana Paula Faria





